Author's response to reviews

Title: Information systems on human resources for health: a global review

Authors:

Patricia L Riley (pyr0@cdc.gov)
Alexandra Zuber (fhq8@cdc.gov)
Stephen M Vindigni (stephen.vindigni@gmail.com)
Neeru Gupta (neeru.gupta@gnb.ca)
Andre Verani (fvi3@cdc.gov)
Nadine L Sunderland (nis9@cdc.gov)
Michael Friedman (mff7@cdc.gov)
Pascal Zurn (zurnp@who.int)
Chijioke Okoro (iff9@cdc.gov)
Heather Patrick (htx7@cdc.gov)
James Campbell (jim.campbell@integrare.es)

Version: 2 Date: 12 September 2011

Author's response to reviews:

Revised Figure 4.